**Author’s response to reviews**

**Title:** Exceptional serological and radiological response to sorafenib in 2 patients with advanced hepatocellular carcinoma and chronic hepatitis C viral infection: Case report and review of the literature.

**Authors:**

Catherine Atkin (catherine.atkin@uhb.nhs.uk)

Philip Earwaker (philip.earwaker@uhb.nhs.uk)

Arvind Pallan (arvind.pallan@uhb.nhs.uk)

Shishir Shetty (shishir.shetty@uhb.nhs.uk)

Pankaj Punia (pankaj.punia@uhb.nhs.uk)

Yuk Ting Ma (y.t.ma@bham.ac.uk)

**Version:** 2  **Date:** 10 Jan 2017

**Author’s response to reviews:**

1. Please provide a list of abbreviations used in the text under the heading "List of abbreviations". The abbreviations should be placed after the Conclusions and before the Declarations.

This has been added

2. Please note that all manuscripts must contain the following sections under the heading 'Declarations':

Ethics approval and consent to participate

Consent for publication

Availability of data and material

Competing interests

Funding

Authors' contributions

Acknowledgements
Authors' information (optional)

The missing sub-sections have now been added

3. Please ensure that all figures referenced in the text are provided when uploading your manuscript. There are figure legends for Figures 1 and 2, but no figures are provided.

This has been uploaded